%
% Fill only lines marked with dots  
% 
%
%
\documentclass[10pt,a4paper,twoside]{article}
\usepackage{latexsym,amssymb}
\usepackage{amssymb,amsmath}
\usepackage{graphicx}
\usepackage{color}
\usepackage{fancyhdr}
\pagestyle{fancy}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%  PLEASE, ONLY IN ENGLISH                                         %
%  PLEASE, ONLY ONE PAGE                                           %
%  PLEASE, USE LATEX2E                                             %
%                                                                  %
%  PLEASE, DO NOT DEFINE NEW COMMANDS                              %
%  USE NEWTHEOREMS BELOW, PLEASE, IF YOU WRITE THEOREMS AND SUCH   % %                                                                  %
%  PLEASE, CONTROL THE SIZE OF YOUR ABSTRACT BEFORE YOU SEND IT    %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
\newtheorem{thm}{Theorem}
\newtheorem{defi}[thm]{Definition}
\newtheorem{axi}[thm]{Axiom}
\newtheorem{clm}[thm]{Claim}
\newtheorem{col}[thm]{Corollary}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{con}[thm]{Conjecture}
\newtheorem{hyp}[thm]{Hypothesis}
\newtheorem{rem}[thm]{Remark}
\newcommand{\proof}{\noindent{\bf Proof:}\hspace{0.2cm}}
\setcounter{thm}{0}
\newcommand{\N}{{\mathbb N}}
\newcommand{\R}{{\mathbb R}}
\newcommand{\W}{W_0^{1,p}(\Omega)}
\newcommand{\eps}{{\varepsilon}}
\newcommand{\ds}{\displaystyle}

\newcommand{\F}{\mathcal F}
\newcommand{\E}{\mathscr E}
\newcommand{\J}{\mathscr{J}}
\newcommand{\Pl}{(\mathscr{P})}
\newcommand{\CC}{\mathbb C}
\newcommand{\into}{\int\limits_{\Omega}}
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\textwidth 15.0cm
\textheight 22.0cm
\oddsidemargin 0.4cm
\evensidemargin 0.4cm
\topmargin  0.0cm

\frenchspacing

\headwidth=\textwidth 	
\fancyhead[HCO]{8th Int. Conf. on Mathematics and Informatics,
	September 9-10, 2021, T\^argu Mure\c s, Romania}
\lhead{}\rhead{}
\fancyfoot{}
%


\begin{document}
\begin{thebibliography}{9} 
% Please, use the following patterns
%
% for articles 

\bibitem{Achalasia}
Lappas B.M., Patel D.A. and Vaezi M. F., \textit{An Overview of Achalasia and Its Subtypes}, PubMed, 2017

\bibitem{Jackhammer}
Sirinawasatien A. and Sakulthongthawin P., \textit{Manometrically jackhammer esophagus with fluoroscopically/endoscopically distal esophageal spasm: a case report}, BMC Gastroenterology, 2021

 \bibitem{Chicago} 
Kahrilas PJ, Bredenoord AJ, Fox M, Gyawali CP, Roman S, Smout AJ, Pandolfino JE; International High Resolution Manometry Working Group. \textit{The Chicago Classification of esophageal motility disorders, v3.0.} Neurogastroenterol Motil. 2015 Feb;27(2):160-74. doi: 10.1111/nmo.12477. Epub 2014 Dec 3. PMID: 25469569; PMCID: PMC4308501.

\bibitem{DCI}
Patel D., Rife C., Grevenitis P., Castell D., \textit{Low Distal Contractile Integral (DCI) on High Resolution Manometry Is Helpful in Predicting Incomplete Bolus Transit}, American Journal of Gastroenterology: October 2012, Volume 107

\bibitem{IRP}
Gong E.J, \textit{Manometrically jackhammer esophagus with fluoroscopically/endoscopically distal esophageal spasm: a case report}, BMC Gastroenterology, 2021

\end{thebibliography}


\begin{center}{\Large\bf
IRP Classification and Probe Positioning Failure Detection in High-Resolution Esophageal Manometry using Machine Learning
}\end{center}
%
\begin{center}{\large\bf\noindent
\textsuperscript{1} Zoltan Czako, Anca Hangan, Gheorghe Sebestyen 

\textsuperscript{2} Teodora Surdea-Blaga, Stefan L. Popa
}\\[2mm]
%%
\textsuperscript{1} Computer Science Department, Technical University of Cluj-Napoca, Romania

\textsuperscript{2} 2nd Medical Department, "Iuliu Hatieganu" University of Medicine and Pharmacy Cluj-Napoca, Romania
%%
\\[1mm]\texttt{
%%	
\{Zoltan.Czako, Anca.Hangan, Gheorghe.Sebestyen\}@cs.utcluj.ro, 
dora\_blaga@yahoo.com, popastefanluci@gmail.com
%%
}\end{center}
%

\vspace*{7mm}

\begin{abstract}

High-Resolution Esophageal Manometry (HRM) is the most used technique for the study of Esophageal Motility Disorders (EMD) like Achalasia type I, type II, type III \cite{Achalasia}, Jackhammer Esophagus \cite{Jackhammer}, etc. The HRM catheter is inserted transnasally and positioned to cover the full length of the esophagus during the procedure. It uses as many as 36 circumferential sensors and generates heat-map-like images, representing the muscle movement in the esophagus during a swallow. These sensors transmits intraluminal pressure data through a high-resolution catheter, which is then translated into dynamic esophageal pressure topography maps. 

Based on these heat-map-like images the Chicago Classification (version 3.0) \cite{Chicago} is applied, which is a formal analytic framework for esophageal motility disorders used to make an esophageal motility diagnosis. The diagnosis of the EDM is a time consuming manual process, doctors needs to make different measurements, like measuring the Integrated Relaxation Pressure (IRP) \cite{IRP}, the Distal Contractile Integral (DCI) \cite{DCI}, etc. on the raw HRM images and then manually apply the Chicago Classification algorithm in order to determine the exact class or type of disorder. Furthermore the esophageal motility diagnosis can be highly affected by the positioning of the catheter, which is the reason why it is critical to make sure that the positioning of the catheter is precise and there is no place for misinterpretation in the generated esophageal pressure topography maps. 

In this paper we present a Machine Learning based solution for detecting probe positioning failures in HRM images, which can be used before applying the Chicago Classification algorithm, this way maximizing the precision of the esophageal motility diagnosis. Furthermore, we created a classifier to automatically determine whether the Integrated Relaxation Pressure is in the normal range or it is higher than the cut-off, based solely on the raw pressure topography images. Determining the IRP type is one of the most important step in the Chicago Classification algorithm, so this work is the first step towards automating the Chicago Classification algorithm using Machine Learning techniques. Automating this algorithm will highly reduce the costs for the hospitals, because the esophageal motility diagnosis will be automatically done, requiring only a nurse to position the catheter, so no specialist will be needed anymore.

In order to detect probe positioning failures and to classify the IRP, firstly we preprocessed the raw HRM images. The first step in preprocessing of the images was to find the region of interest, by finding the part of the image which represents the exact moment of swallowing. After this step we resized the images and we rescaled the pixel values so they can be used as input for Deep Learning models. After preprocessing the images we tried multiple Deep Learning models to classify the images as normal or abnormal images (abnormal means wrong catheter position) and to determine the exact class of the IRP. The result of the trained Neural Networks (NN) were very promising, with an accuracy and f1-score over 95\% for most of the NN models that we tried. The results of these experiments are presented at the end of our article.
    
\end{abstract}


%

\end{document}


